Effects of the Invasive Freshwater Mussel *Limnoperna fortunei* on Sediment Properties and Accumulation Rates

Running title: Effects of Limnoperna fortunei on sediments

Romina Tokumon^{1, 2}, Demetrio Boltovskoy¹, and Daniel Cataldo^{1, 2}

 ¹ Instituto de Ecología, Genética y Evolución de Buenos Aires, Universidad de Buenos Aires-CONICET, Intendente Güiraldes 2160, Buenos Aires C1428EGA, Argentina;
 ² Museo Argentino de Ciencias Naturales "Bernardino Rivadavia"-CONICET, Av. Angel Gallardo 470, Buenos Aires C1405DJR, Argentina

Corresponding author: Demetrio Boltovskoy (demetrio@ege.fcen.uba.ar)

Key points:

- The invasive freshwater mussel *Limnoperna fortunei* strongly enhances sediment buildup rates and their contents in organic matter and total nitrogen
- Ecosystem-wide effects of these modifications are likely very significant, especially for soft-bottom, infaunal organisms

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2018JG004399

Abstract Since its introduction into South America around 1990, the freshwater bivalve Limnoperna fortunei (the golden mussel) has spread rapidly, and is now a dominant component of the benthic and periphytic fauna in many rivers, lakes and reservoirs. Sizable impacts of this non-indigenous species on nutrient recycling, plankton abundance and composition, and trophic relationships with fishes have been reported, but its effects on the sediments have received little attention. In this work, we use eighteen 20 L flow-through experimental units with and without mussels where changes in the mass and characteristics of the sediments accumulated throughout a yearly cycle in monthly, biannual and annual intervals are analyzed. Experimental units with mussels yielded almost two times more sediments than units without mussels, and contained significantly higher loads of organic matter (OM) and total N. Total P was not affected by the presence of mussels. Sediments accumulated in the biannual and annual experimental units agreed well with the yields of the monthly units, but the vertical stratification of organic matter, N and P was unpatterned. Seasonal changes in the volume of total sediments, biodeposits, and their OM and N contents, were positively associated with ambient water temperature and with intermediate (~150-250 NTU) turbidity. Our results suggest that ecosystem-wide modifications in the living conditions of the benthic epi- and infaunal organisms in waterbodies invaded by the mussel are likely significant, although variable locally, regionally and across taxa.

Accepte

Plain language summary In the last decades, thousands of plants and animals have spread into new marine, freshwater and terrestrial habitats through human-mediated activities. Worldwide, freshwater ecosystems seem to have suffered the most. The Asian golden mussel, a small (~2-3 cm) bivalve mollusc, was introduced in South America around 1990, and by 2017 it was present, often at densities over 200,000 individuals per square meter, in many rivers, lakes and reservoirs throughout five countries. The mussel, which lives attached to hard substrates, is presently a major biofouling nuisance for many industrial installations, and has been shown to affect pre-existing biological interactions significantly. In this work we show that, in the estuary of the Río de la Plata (South America), the golden mussel doubles the rates of sedimentation of suspended particles, and strongly enhances their contents in organic matter and nitrogen. These effects are likely to increase the rates of accretion along the Río de la Plata estuary front, as well as to modify the makeup of the communities that dwell in the sediments. The golden mussel is expected to continue expanding northwards, and its impacts will likely affect many of the new waterbodies colonized.

Keywords: Limnoperna fortunei, Golden mussel, impact, sediments, organic matter, Nitrogen, Phosphorus

1. Introduction

The presence of filter-feeding species has long been recognized as a key factor that strongly influences the accumulation rates and characteristics of sediments (Dame & Olenin, 2005; Jørgensen, 1990). However, while the effects of indigenous, natural filter-feeders are the result of interactions that evolved throughout thousands or millions of years, those of species cultured (shellfish farms) or introduced by man (non-indigenous species) may disrupt pre-existing conditions over very short time-scales (Nalepa & Schloesser, 2014; Shumway, 2011). In this respect, bivalves are particularly important, both because they are the object of extensive commercial farming worldwide (Shumway, 2011), and because they include a large proportion of the invasive invertebrates (Karatayev et al., 2009).

Several previous surveys investigated the effects of various filter-feeding molluscs on the sediments, including native and introduced species, as well as cultured bivalves. While there is general agreement that total sediments, organic matter and N are enhanced in the presence of the mussels, the behavior of P, the proportions and long-term fate of these biodeposited sediments, and their ultimate effects on bottom properties and buildup rates over large areas are controversial (Crawford et al., 2003; Gergs et al., 2009; Roditi et al., 1997; Ruginis et al., 2014; Shumway, 2011; Zaiko et al., 2010).

During the last ~60 years Limnoperna fortunei (the golden mussel), native to southern China, has been introduced in Hong-Kong, Korea, Taiwan, Japan and South America (Boltovskoy & Correa, 2015). In South America, where it was first recorded in the Río de la Plata Estuary around 1990, it spread rapidly upstream the Paraná and Uruguay rivers, and is presently a dominant component of the benthic and epiphytic fauna in two major watersheds (Río de la Plata and São Francisco) and several smaller ones throughout five countries (Argentina, Bolivia, Brazil, Paraguay, and Uruguay) (Barbosa et al., 2016; Oliveira et al., 2015). The introduction of the golden mussel has become an important biofouling nuisance for industrial plants that use freshwater from the colonized waterbodies (Boltovskoy et al., 2015a), and is reshaping biological interactions. Investigation of the impacts of this mussel on the environment and on other organisms has chiefly centered on nutrient recycling, plankton abundance and composition, and trophic relationships with fishes (Boltovskoy, 2015). Effects on the invertebrates associated with its colonies were assessed comparing the fauna present in mussel beds or druses, to nearby areas of the same substrates or sites barren of mussels (Karatayev et al., 2010; Sardiña et al., 2011; Sylvester & Sardiña, 2015), but no previous research focused on the changes that L. fortunei beds engender in the sediments.

We address five questions in this survey. (1) How much of the sediments and sestonic organic matter (OM), N and P settling to the bottom is associated with the mussels' presence? (2) Which are the ambient conditions that modulate this process? (3) How much OM, N and P is sequestered in the sediments? (4) How do these materials behave in the sediments (up to) one year after burial? and (5) Which are the potential effects of the mussel-modified sediments on the benthic biota?

2. Material and methods 2.1 Regional setting

This work was conducted on the premises of the water treatment plant AySA (Agua y Saneamientos Argentinos Sociedad Anónima), in Buenos Aires (Argentina), which provided a permanent raw water supply to our experimental device (Experimental Units, or EU henceforth; Figures 1, 2). The plant processes water from the Río de la Plata Estuary, which is gravitationally fed through a 5 m diameter pipe from an intake located ~1 km offshore (34.54°S, 58.42°W). At this location the water is fresh year-round (<3 PSU - Practical Salinity Units) and contains high loads of suspended solids (SS; ~100 mg L⁻¹) (Bazan & Arraga, 1993). The estuary, with a catchment area of 3.1 million km⁻² and a mean discharge of ~25,000 m⁻³ s⁻¹, is chiefly fed by two major rivers: the Paraná River (~74% of the input) and the Uruguay River (~23%) (Framiñan et al., 1999). Seasonally, discharge values are normally highest in March-June, which usually coincides with peaks in total SS, but SS may also increase after peak discharge periods, when flood water leaves the inundated flood plain (Depetris et al., 2005), as well as during periods of strong winds, which result in vertical turbulence and resuspension of bottom deposits (Moreira et al., 2013).

2.2 Experimental setup and sample analyses

The experimental design consisted of eighteen 20 L EU enclosed in a box covered with a removable lid (Figure 1). Each EU received a constant supply of estuary water at a rate of ~0.8-0.9 m³ d⁻¹ (~556-625 mL min⁻¹) from a head tank by gravity, which overflew through a nozzle at the top of each EU (Figure 2). Half of the EU contained a meshwork basket suspended below the water surface (Figure 2a) with 87-255 (mean: 199) adult (>98% between 8 and 28 mm in length) mussels collected ~2-3 days before from nearby locations along the coast of the estuary and rinsed with tap water (EU with mussels, or +M henceforth). Baskets were placed near the water surface (rather than close to the bottom) in order to avoid their burial due to sediment accumulation, especially in the biannual and annual units. The resulting mussel densities were 1318-3864 (mean: 3015) ind. m⁻², which

are roughly within the range of those observed in nature (see Discussion below). The other half of the EU were barren of mussels (controls; CTR henceforth). In order to minimize disturbing and injuring the animals, clumps of mussels of approximately equal sizes were used (rather than isolated individuals). The size of the mussels varied between periods (Table S1 in the supporting information), for which reason our results were normalized to total mussel tissue dry weight (MTDW). Nevertheless, filtration rates and the rates of production of feces and pseudofeces are not homogeneous across sizes (Sylvester et al., 2005), which may involve some ambiguities in our estimates. On the other hand, because the experimental animals used were collected in the field shortly before each experimental period, their sizes are representative of the dominant population size-structure at that particular time of the year and water temperature, mimicking natural conditions better than if animals more similar in size were used. Upon termination of each monthly experimental period, animals were removed, counted, measured, and their MTDW was estimated using the size to dry weight relationships proposed by Sylvester et al. (2007a):

Mussels <10 mm: MTDW (g)= $0.000119 \times e^{0.416 \times \text{mm shell length}}$ Mussels >10 mm: MTDW (g) = $0.00396 \times e^{0.113 \times \text{mm shell length}}$

The experiment was deployed on 4 Dec 2014 and terminated 362 days later, on 1 Dec 2015, yielding a total of 66 monthly observations: eleven 25-35 d periods (the 19 d interval between 15 Oct and 3 Nov 2015 was lost due to technical problems), 12 biannual, and 6 annual (Figure S1 in the supporting information).

Of the 18 EU, six (3 +M and 3 CTR) were used for monitoring monthly (25-35 d) changes, six had a biannual (173 and 189 d) duration, and six were maintained throughout the entire experimental period (362 d) (Figure S1 and Table S1 in the supporting information). In the biannual and the annual +M, mussels were replaced every two months to ensure their vitality, minimize the retention of dead organisms, and following the dominant size-structure in nature. Their number, size and biomass were estimated as described above on each replacement event and at termination. Each period (monthly, biannual and annual) and condition (with and without mussels) was repeated simultaneously in triplicate in randomly assigned EU. Mortality in the +M was extremely low, never exceeding ~3% of the animals. These dead mussels were excluded from our estimates, which may have introduced a small bias because they might have been represented by dead individuals retained within the clumps of living ones at the start of each period, or by individuals that died during the experiment.

Water flow into each EU was monitored throughout the experiment. In total, 710 measurements of water flow were performed, both on and between sampling dates. Differences between the 3 +M and the 3 CTR replicates of each period were generally <15%. Differences between successive periods were higher (up to 35%, mean: 13%), for which reason total sediment volumes were adjusted to account for these variations (see below and Table S1 in the supporting information).

Water properties were measured at the intake at the start and end date of each monthly experiment (Figure S1 in the supporting information), as well as at irregular intervals in between: temperature ($^{\circ}$ C), pH, conductivity (μ S s⁻¹), dissolved oxygen (mg L⁻¹). Turbidity (Nephelometric Turbidity Units - NTU) was assessed at the intake into the device and at the outlet of each EU. For temperature, in addition to the periodic measurements, we also recorded the values at 30 min intervals by means of miniature temperature, battery-powered data loggers ("thermo buttons"; model 25L, Proges Plus, Willems, France), positioned either at mid-depth (two EU), or in the vicinity of the surface and next to the bottom (one EU) (Figure 2).

At the end of each monthly period, all sediments accumulated in the monthly EU were recovered and subsampled for assessment of their dry weight (60°C until constant weight), organic matter (OM) contents (in %, loss on ignition: difference between dry weight and weight after igniting for 3 h at 500°C). Total P (ascorbic acid method) and total N (cadmium reduction method) in the sediments were assessed spectrophotometrically after digestion with potassium persulfate (Valderrama, 1981). For the biannual and annual EU, total dry weight was estimated as above, but %OM, and total P and N were assessed at 2 cm intervals using cores 2.5 cm in diameter obtained from the EU (3-9 cores from each EU; Figure 2). Except where noted otherwise, for the biannual and annual EU, %OM, P and N are based on the mean values of all 2 cm intervals in the corresponding cores.

Because experimental periods, mussel biomass, and the volumes of water supplied to the EU were not identical, total sediments accumulated in the +M were normalized to 5.2 g of MTDW (the average for all experiments), one day, and 1 m³ of water per day, and their properties (%OM, P, N) to 5.2 g of MTDW. In the CTR, the mass of sediments was normalized to one day and 1 m³ of water per day.

This experimental design allowed differentiating passively settling sediments (in the CTR; "passive sediments" henceforth), from passive + biologically-mediated sedimentation ("passive+biodeposited sediments" henceforth), and, by subtraction, sediments accumulated due to the production of feces and pseudofeces by the mussels ("biodeposited sediments" henceforth). Differences in sedimentation rates between CTR and +M are unlikely to have been affected by hydrodynamic effects due to the presence of the cages (EUM) because the cages occupied only ~1% of the total volume of the EU.

SS in the water (~100 mg L⁻¹, with peaks >600 mg L⁻¹; Simionato et al., 2011) were not assessed in our work, but in the Río de la Plata total SS are tightly coupled with NTU (total mg L⁻¹SS=0.73 NTU, R=0.93; Moreira et al., 2013); we therefore used this conversion factor when comparing or results with literature data where total SS were informed.

Analyses of relationships between environmental variables and experimental results were based on bivariate methods (correlations, t-tests). Multivariate treatments were hindered by the strong collinearity between several of the independent variables considered (e.g., O2, water temperature, and mean mussel size; conductivity and turbidity). ANOVA or Kruskal Wallis-Mann Whitney were used to compare mean differences in the response variables (total and biodeposited sediments, %OM, P, N) for the monthly periods with different temperature and turbidity offsets. Differences in turbidity at the outlet of the EU as compared with those at the intake were assessed with ANOVA and ANCOVA. Parametric methods were used when raw or transformed data met the assumptions required. When test assumption were not met, the statistical indicators were based on ranked values (nonparametric) (program PAST, Hammer et al., 2001).

3. Results

3.1 Environmental conditions

Water temperature followed the usual seasonal trend for the upper Río de la Plata Estuary, with highest values (~25-28°C) in December-March, and lowest (~12-16°C in June-September (Figure S2 a in the supporting information). In the EU, temperature varied little with depth. The EU provided with two temperature data loggers (one near the surface and one next to the bottom, Figure 2) yielded a mean daily difference of 0.32°C (SD=0.33). Daynight differences in temperature were also relatively small (mean for the entire experimental period ~0.7°C), with highest values (~1-2°C) in the warmest months, and lowest (~0.6°C) during the colder period (Figure S3 in the supporting information).

Dissolved oxygen was always high (>6 mg L^{-1}) and close to saturation (75 to >100%, mean: 90%). Variations in pH were small (generally 6.5 to 7.5; Figure S2 b in the supporting

information). Conductivity and turbidity depicted similar trends, with highest values in the fallwinter (April-June; Figure S2 c in the supporting information).

Temperature was highly correlated with O_2 (Spearman's R=-0.895, P<0.001), which simply reflects the fact that oxygen is more soluble in colder waters; and with mussel length (Spearman's R=-0.691, P=0.019). *L. fortunei* reproduces during the warmer months and therefore high water temperatures are associated with higher proportions of juveniles and lower mean mussel size. Conductivity generally paralleled turbidity (Figure S2 c in the supporting information), but the correlation was not significant (Spearman's R=0.509, P=0.110). We therefore centered our attention on the two independent variables that were not associated throughout the experimental period (water temperature and turbidity; Spearman's R=0.067, P=0.100, N=613), and that have been shown in previous surveys to affect mussel filtration rates most clearly (Boltovskoy et al., 2015b; Tokumon et al., 2015).

3.2 Sediment accumulation rates

Sediment accumulation varied greatly throughout the year (Figure 3a, 4a, b). Mussels significantly increased sedimentation (Mann-Whitney, P<0.001), where mean +M rate was 40.5 g DW d⁻¹, relative to the CTR mean of 21.6 g DW d⁻¹ (Figure 3a, Table S1 in the supporting information). On average for all the experimental periods, biodeposited sediments (i.e., +M minus CTR) represented 47% (range: 7-65%) of the totals (i.e., passive+biodeposited sediments).

Turbidity at the inlet and outlet of all EU were correlated, but the slopes of the +M and CTR were significantly different (ANCOVA, p<0.011). CTR aligned closer to the 1:1 ratio than +M (Figure S4 a in the supporting information). In the CTR, the outlet was 74% of the inlet value, while the +M outlet was significantly lower at 62% of the inlet (Mann-Whitney P<0.001; Figure S4 b in the supporting information).

Although monthly values for the +M paralleled those in the CTR (Figure 5a), in the former they were significantly associated with temperature, whereas in the CTR they were not (Figure 4a). Ambient turbidity did not affect sediment accumulation rates significantly..

3.3 Biodeposited sediments

In the +M, biodeposited sediments were clearly (positively) associated with temperature (Figure 4c). Association with turbidity was also significant, but highest values coincided with

periods when ambient turbidity was ~150-250 NTU, decreasing below and above these values (Figure 4d).

3.4 Organic matter (OM)

Proportions of OM were almost invariably higher in the +M than in the CTR (Figure 3b). On average for all the EU, %OM in the CTR was 69% of that in the +M. Differences between +M and CTR were statistically significant (mean for +M: 4.8%, CTR: 3.3%, P=0.011, t-test). The monthly means were generally similar with those recorded in the annual EU (Figure 3b).

Although mean differences at temperatures above and below 20°C differed more in the +M than in CTR, neither set of monthly units (+M and CTR) showed significant associations between %OM and water temperature (Figure 4e).

In the CTR, ambient turbidity variations had no effect on %OM, whereas in the +M %OM tended to increase with increasing turbidity at the intake up to values ~150-250 NTU, dropping thereafter (Figure 4f). In the biannual and annual EU, the vertical distribution of %OM down to 6-8 cm (CTR), and to 12-18 cm (+M) was homogeneous and showed no trend (Figure 6a-c).

3.5 Total P

Total P differed little between +M and CTR (Figure 7a) (mean for +M=0.215, mean for CTR=0.206 mg g⁻¹ DW; Man-Whitney test P=0.482), although month-to-month variability was significantly higher in the +M (SD=0.051) than in the CTR (SD=0.026) (F test P=0.044). In the +M, total P was positively associated with temperature, but not in the CTR, where the correlation was non-significant (Figure 4g).

Associations with turbidity were marginally significant (negative) in the CTR, and nonsignificant in the +M (Figure 4h). Monthly variations in +M and CTR were not associated (Figure 5c). In agreement with the monthly results, in the biannual and annual EU the vertical distribution of total P varied randomly between +M and CTR, as well as with depth (Figure 6d-f).

3.6 Total N

N concentrations were higher in the +M than in the CTR in all the monthly EU (Figure 7b), as well as at all depths in the cores retrieved from the biannual and annual EU (Figure 6g-i).

Monthly means differed significantly (P=0.002, Mann-Whitney test), but the temporal trends in the +M and the CTR clearly paralleled each other (Figure 5d). In the +M (but not in the CTR), total N was positively correlated with temperature (Figure 4i). Cores from the biannual and annual EU showed no vertical pattern (Figure 6g-i).

4. Discussion

4.1 Sediment accumulation rates

Our results support the notion that *L. fortunei* has a major effect on sedimentation rates and the properties of the resulting deposits (Figures 3, 7), and that strongest enhancements of biodeposited sediments occur at times of higher water temperatures (Figure 4). The presence of mussels almost doubled sediment accumulation rates, and significantly increased their contents of OM and total N.

Because of differences between our experimental settings and natural conditions, the sediment accumulation rates in our EU are difficult to compare with field data. However, a rough estimate suggests that our values are a reasonable proxy of actual sedimentation rates in the estuary. Mean sediment accumulation rates in the +M were 3.5 cm m⁻² y⁻¹, whereas in the CTR they averaged 1.9 cm m⁻² y⁻¹. These values are comparable with those estimated using cores from the upper Río de la Plata Estuary dated with ²¹⁰PB and ¹³⁷Cs: around 0.25-2.6 cm y⁻¹ (Bonachea et al., 2010; Di Gregorio et al., 2007; Schuerch et al., 2016), and up to over 5 cm y⁻¹ in some near-shore areas (Colombo et al., 2007). A rough approximation comparing historical (since 1778) and present-day elevation data for the mouth of the Paraná River Delta (Sarubbi, 2007) suggests vertical growth rates around 1-3 cm y⁻¹.

The amount of sediments accumulated in the biannual and annual EU was somewhat lower than the sum of the corresponding monthly EU. Aside from the circumstance that the monthly EU missed a 19 days interval (Figure S1 in the supporting information), this difference is probably due to the fact that in the biannual and annual EU the top of the sediment layer was closer to the outlet, and therefore a larger proportion of the incoming particles were flushed out. However, this effect does not seem to have affected comparisons between the monthly EU, where the top sediment layer was much farther away from the outlet (\sim 25-27 cm) than in the biannual and annual ones (\sim 5-10 cm), and the ratio of turbidity at the intake vs. turbidity at the outlet was uncorrelated with sediment buildup rates (+M, Spearman's R=-0.367, P=0.336; CTR, Spearman's R=-0.409, P=0.212).

Comparison of our results with previous data on contrasts between passive and passive+biodeposited sedimentation rates show roughly comparable values. On the basis of four 24 h experiments conducted in chambers with and without zebra mussels (Dreissena polymorpha) in the Hudson River, Roditi et al. (1997) estimated a mean rate of biodeposited sediments of ~55 mg DW mussel⁻¹ d⁻¹, whereas our values were ~97 mg DW mussel⁻¹ d⁻¹. The mean size of the mussels used in the two studies was very similar (~17 mm), but the amount of SS is almost ten times lower in the Hudson River (~10 mg DW L⁻¹), than in the Río de la Plata Estuary (~100 mg DW L⁻¹). Suspended solids and species-specific biofiltration differences likely account for our higher values. Previous studies with zebra mussels support this assumption: at 50-87 mg DW SS L⁻¹ biodeposition rates of 100-164 mg DW mussel⁻¹ d⁻¹ were reported (Klerks et al., 1996; Reeders & bij de Vaate, 1992). Mussel pumping rates and the production of feces and pseudofeces vary in response to the concentrations of SS in the water, with both positive (Gergs et al., 2009; MacIsaac & Rocha, 1995; Nalepa et al., 1991) and negative (Sprung & Rose, 1988) relationships having been reported, but the range of inorganic SS loads is probably a major factor to which the mussels respond bimodally. For L. fortunei, Tokumon et al. (2015) found that filtration rates increase between 0 and 100 mg DW SS L⁻¹, dropping sharply at values above 500 mg DW SS L⁻¹. In the Salto Grande Reservoir (Argentina-Uruguay), at values around 5-15 mg DW SS L⁻¹, Cataldo et al. (2012b) found that, after 35 d, 400 L mesocosms with L. fortunei accumulated 5-6 times more sediments than mesocosms without mussels, which is higher than our values with Río de la Plata water (~2), where SS were up to >270 mg L^{-1} (Table S1 in the supporting information). In our +M, biodeposited sediments showed an increasing trend up to ambient turbidity values ~150-250 NTU (~110-180 mg DW SS L⁻¹, Moreira et al., 2016), decreasing at higher values (Figure 4d), which is in line with the results of Tokumon et al. (2015).

4.2 Organic matter and total N

With one exception, %OM and N were always significantly higher in the presence of mussels (+M) than in the controls (CTR) (Figures 3, 7b). It should be noted that both %OM and N in all the EU were influenced by ambient values as well; although values in the +M were consistently higher than those in the CTR, monthly changes in the EU with and without mussels paralleled each other (Figure 5b, d). Values for %OM in the CTR were closely comparable with information for the surface sediments of the Río de la Plata Estuary (Moreira et al., 2016; Ronco et al., 2008). Our results agree with previous data where biodeposited sediments were found to be highly enriched with OM and/or N when compared with passively deposited sediments (e.g., Graf & Rosenberg, 1997; Hatcher et al., 1994;

Jordan, 1987; Kaspar et al., 1985; Roditi et al., 1997). Bivalve biodeposits can enrich sediment OM and impact N cycling in sediments through increasing direct denitrification (i.e., denitrification of water column nitrate), reducing coupled nitrification-denitrification, or increasing ammonium efflux, depending on overarching factors such as eutrophication and habitat heterogeneity (Smyth et al., 2015).

4.3 Total P

As opposed to OM and N, the concentrations of total P were not higher in the +M than in the CTR (Figure 7a). While the effects of biodeposits on the enrichment of sediments with OM and N were observed in many surveys (see above), in particular those aimed at assessing the environmental effects of shellfish cultures in marine coastal areas (Shumway, 2011), results with P are more controversial (Biswas et al., 2009; Nalepa et al., 1991; Strayer, 2014; Wetzel, 2001). Several studies concluded that biodeposits contain significantly higher loads of P than seston and/or passive sediments (Jordan, 1987; Kleeberg, 2002; Mosley & Bootsma, 2015; Nalepa et al., 1991; Sornin et al., 1986), and that P concentrations in the water column decrease after colonization of the waterbody with filter-feeding mussels (Mayer et al., 2014; Mida et al., 2010). In contrast, many others found no differences in P concentrations in sediments influenced by biodeposited matter with those farther away from biodeposit-producing organisms (Carlsson et al., 2009; Hatcher et al., 1994; Mellina et al., 1995; Zaiko et al., 2010), recorded strong increases in phosphates in the water column in the presence of mussels (Boltovskoy et al., 2009; Cataldo et al., 2012a; Cataldo et al., 2012b), and no association between OM and P in sediments (Tye et al., 2016). These dissimilar results may be related with the lower sediment retention of P than of N (Dillon et al., 1990). Seasonal changes in P loads in the sediments have been recorded in some studies (Mosley & Bootsma, 2015), but not in others (Mazouni et al., 1996; Nalepa et al., 1991). In our data no seasonal trends were found, which may suggest that proportions of P in the water are seasonally more stable than those of OM and N, and/or that biodeposited P is largely released to the water column.

Several mechanisms have been suggested to account for this different behavior of P (as compared with OM and N), including the release of unbound P due to OM-induced oxygen depletion (Newell, 2004; Newell et al., 2005; Shumway, 2011), bioturbation by infaunal organisms, in particular chironomids and mayflies, that can reduce P retention in the sediments and favor its release as bioavailable P into the water column (Biswas et al., 2009; Chaffin & Kane, 2010), differences in the functions of P and N in animal physiology (Vanderploeg et al., 2017), and in the physical and chemical traits of the water and the

sediments (Li et al., 2013; Reynolds & Davies, 2001; Tye et al., 2016). It has been suggested that sediment retention of P bound to the dead shells (rather than to the soft tissue and the biodeposits) of the mussels can take up to 100 y to reach a balance before sediments become a net source of P, whereas until that point sediments are a net sink (Ozersky et al., 2015). However, in the Ca-poor South American waterbodies (Karatayev et al., 2007) empty *L. fortunei* shells have never been observed to accumulate in the sediments, and their dissolution can be even faster than the disintegration of the soft tissue (Boltovskoy et al., 2009).

4.4 Vertical distribution of properties in the sediments

In the cores obtained from the biannual and annual EU, contrasts between +M and CTR were similar to those observed in the monthly EU, but neither of the variables measured showed consistent vertical patterns (Figure 6). To compare our CTR data for OM with field conditions, we took several 20-30 cm cores in the vicinity of the Río de la Plata Estuary (Pajarito River, Paraná delta, ~5 km upstream from the estuary), and assessed their OM contents in 1 cm intervals. Highest %OM figures (>4%) were recorded at 1 and 12 cm, but down to 26 cm no trend was observed (mean=3.5%, range=2.7-4.3%, Spearman's R of depth in the core vs. %OM: -0.316, P=0.116). In sediments of the upper Río de la Plata Estuary, organic C in the sediments has been reported to drop from ~3% at the surface (5% OM, according to the conversion factor proposed by Hoogsteen et al., 2015) to ~2% (3.3% OM) at 4 cm, oscillating around 2% down to 30-50 cm (Schuerch et al., 2016), which agrees well with our data (mean for +M: 4.8%, CTR: 3.3%; Figures 3, 6, and Table S1 in the supporting information). Previous literature information from freshwater bodies elsewhere shows widely dissimilar results (e.g., Bloesch, 2004; Grenz et al., 1990; Sobczyński & Joniak, 2009; Torres et al., 2012), suggesting that local spatial and temporal conditions result in dissimilar vertical profiles for these attributes.

It should be borne in mind that our survey covered 1 year only, whereas remineralization of the OM can take longer than that, and therefore the proportions of OM ultimately buried as refractory carbon, and especially N, can be lower (Bloesch, 2004). Nevertheless, the fact that the OM contents of our EU compare favorably with downcore field data suggests that our experimental results are not unrealistic.

5 Considerations for scaling up results to predict local and regional impacts

The effects of *L. fortunei* on the sediments are similar to those reported for the effects of the zebra mussel and other filter-feeding molluscs (Klerks et al., 1996; McCall et al., 1986; Mellina et al., 1995; Roditi et al., 1997; Ruginis et al., 2014; Zaiko et al., 2010), but

extrapolation of our experimental conditions to the field is not unequivocal. Among the most important caveats are: (1) The lack of adequate data on densities of *L. fortunei* over large areas; (2) Uncertainties associated with the ease of resuspension of the passive+biodeposited sediments as compared with that of passive sediments; and (3) Differences in the winnowing and dispersion of the biodeposits in different lotic and lentic waterbodies colonized by the mussel.

Data on *L. fortunei* densities over large areas are extremely scarce. Boltovskoy et al. (2009) assessed *L. fortunei* densities in the reservoir Embalse de Río Tercero (Argentina), reporting figures of up to over 6200 ind. m⁻² (mean for the entire reservoir: 959 ind. m⁻²). The extensive floodplains of many lowland rivers colonized by the mussel in South America (Oliveira et al., 2015) are densely populated by the water hyacinth *Eichhornia crassipes*, whose roots host up to >2760 ind. m⁻² (Musin et al., 2015) (in our EU, mussel densities were ~3000 ind. m⁻²). The rate of production of biodeposits depends on both mussel densities and on the volume of water available to the mussels for filtration. In Embalse de Río Tercero, the average volume of reservoir water per mussel per day is theoretically around 0.4 L (Boltovskoy et al., 2009). In our experiment water was supplied to the EUM at ~6 L mussel⁻¹ d⁻¹, which is higher than the above value, and most probably largely in excess of the volumes that *L. fortunei* can process (<2 L ind.⁻¹ d⁻¹; Boltovskoy et al., 2015b). We therefore anticipate that, in this respect, the experimental conditions used are generally within the range of actual conditions in many South American waterbodies.

Differences in the resuspension rates, winnowing and lateral advection between biodeposited and passively deposited sediments are controversial. Strayer (2014) suggested the mussel biodeposits are heavier than passive sediments and therefore may have a higher tendency to be retained locally, yet other authors found the opposite (McCall et al., 1986; Roditi et al., 1997).

Our results indicate that passive+biodeposited sediments are enriched with smaller-sized particles as compared with passive sediments. The decay in turbidity of untreated, well mixed (all depths) resuspended sediments recovered from the annual EU shows that after 2 h CTR (passive sediments) are ~70% less turbid than passive+biodeposited sediments (+M), indicating that they contain higher proportions of larger, rapidly settling particles (Figure S5 in the supporting information). Indeed, in the passive+biodeposited sediments (+M) the proportions of clay (<4 μ m) are two times higher than in the passive sediments (CTR), whereas those of silt (4-60 μ m) are ~30% lower (Figure S6 in the supporting information). These differences between the +M and CTR are very highly significant for both

clay and silt (P<0.001, Mann-Whitney tests). These contrasts are most probably due to the higher retention of clay in the mussel's mucus-bound feces and pseudofeces, and therefore are conceivably less prone to resuspension. The fate of mussel biodeposits is likely variable. Resuspension may relocate these OM-rich materials and the inorganic particles bound in the feces and pseudofeces, which subsequently settle in lower energy areas; it may disperse them facilitating downstream transport and exportation; or simply return them to the water column where they are ultimately degraded. The balance between these processes is obviously waterbody and site-specific (Boyd & Heasman, 1998; Crawford et al., 2003; Hartstein & Rowden, 2004; Strayer, 2014). In rivers, downstream dispersion probably prevails over local accumulation, whereas in lakes and reservoirs, where currents are weaker, sediments are usually transported from the shallow coastal belt to deeper areas. In the reservoir Embalse de Río Tercero, >98% of the mussel beds are located along the coasts at depths <10 m, which is where hard substrates are available (Boltovskoy et al., 2009). Yet, eight years after L. fortunei invaded this reservoir, the coastal rocky fringe was still practically free of sediments, except for a thin layer of loose, highly mobile particles on top of the mussels (Boltovskoy et al., 2009). Thus, rather than accumulating above the mussel beds, the biodeposits produced by these colonies are removed from the coastal belt (where turbulence due to wave action is highest) and settle elsewhere, at depths >10 m, which are almost completely dominated by silt (Boltovskoy et al., 2009). In the Río de la Plata Estuary resuspension is very high, especially in the upper freshwater section where water depths are lowest. However, this section, as well as much of its coasts farther downstream, are densely populated by rooted macrophytes (Scirpus spp.), whose growth is probably facilitated by the enhanced nutrient levels in the sediments (Bertness, 1984), and which have been shown to decrease sediment resuspension rates significantly (Horppila & Nurminen, 2005).

Most surveys on the effects of *L. fortunei* on the benthic fauna show that invertebrates that dwell among the mussels are more diverse and abundant than those outside of the mussel colonies (Burlakova et al., 2012; Karatayev et al., 2010; Sylvester et al., 2007b). These differences have been attributed to the enhanced structural complexity represented by the mussel beds and to the mussels' biological activity (Sardiña et al., 2008). On the other hand, studies of bare sediments influenced vs. uninfluenced by the presence of mussels are very scarce and their results are controversial. In a study of soft-bottom communities at different distances from large *L. fortunei* beds, Sardiña et al. (2011) concluded that, at a pristine site, species richness and abundance increased away from the mussels, but at a polluted one the trend was irregular. In several Brazilian reservoirs, areas populated by the non-indigenous bivalves *L. fortunei* and *Corbicula fluminea* were reported to host lower overall diversity and

abundance than areas barren of them, yet the biomass of the non-indigenous bivalves was uncorrelated with that of the native fauna (Pearson's R=0.044) (Linares et al., 2017).

The effect of sediment changes on the biota is unresolved, but is likely to be significant and highly variable among locations and taxa. Higher OM in biodeposits, as well as their higher proportions of small-sized particles, may reduce O_2 levels in sediment pore-water (Yamada et al., 2012). However, strong and persistent near-bottom oxygen depletion events are unlikely in the Río de la Plata Estuary because wind-driven vertical mixing of its shallow (mostly <10 m) water column and massive sediment resuspension events are very common (every 12.5 h to 8.5 d; Moreira et al., 2013). Major changes in benthic communities are associated with the percent of time under hypoxic conditions, which is most probably low in the estuary, rather than with instantaneous, short-lived low oxygen events (Cicchetti et al., 2006). Thus, we anticipate that, in the estuary, the facilitating effects of the OM-rich sediments on deposit-feeding organisms outweigh the inhibiting ones of lower dissolved O_2 concentrations. Facilitating effects are probably more marked for motile animals, like fishes, than can move away from the occasionally oxygen-depleted layer. Epi- and infaunal invertebrates permanently restricted to the sediments and highly sensitive to short periods of hypoxia can conceivably suffer a negative impact.

Accepte

Acknowledgments

Data are available in the supporting information and at http://doi.org/10.5281/zenodo.1233327. This work was supported by grants from the Agencia Nacional de Promoción Científica y Tecnológica (Argentina) PICT 2012 0070 and PICT 2015 2598, and from the Consejo Nacional de Investigaciones Científicas y Técnicas (CONICET, Argentina) PIP 112 201101 00564 to DB. We are grateful to Agua y Saneamientos Argentinos S.A. (AySA) for providing the facilities, data, and the water supply for this survey. RT was supported by a CONICET doctoral fellowship. The comments of the editor and two anonymous reviewers on an earlier version of this work are greatly appreciated.

Acc

References

- Barbosa, N. P. U., Silva, F. A., Oliveira, M. D., Neto, M. A. S., De Carvalho, M. D., & Cardoso, A. V. (2016). *Limnoperna fortunei* (Dunker, 1857) (Mollusca, Bivalvia, Mytilidae): first record in the São Francisco River basin, Brazil, *Check List, 12*(1), 1846. https://doi.org/10.15560/12.1.1846
- Bazan, J. M., & Arraga, E. (1993). El Río de la Plata, un sistema fluvio-marítimo frágil?:
 Acercamiento a una definición de la calidad de sus aguas. In A. Boltovskoy & H. L.
 López (Eds.), *Conferencias de Limnología* (pp. 71-82). La Plata (Argentina): Instituto de Limnologia "Dr. R.A. Ringuelet".
- Bertness, M. D. (1984). Ribbed mussels and *Spartina alterniflora* production in a New England salt marsh, *Ecology*, *65*(6), 1794-1807. https://doi.org/10.2307/1937776
- Biswas, J. K., Rana, S., Bhakta, J. N., & Jana, B. B. (2009). Bioturbation potential of chironomid larvae for the sediment–water phosphorus exchange in simulated pond systems of varied nutrient enrichment, *Ecological Engineering*, *35*(10), 1444-1453. https://doi.org/10.1016/j.ecoleng.2009.06.004
- Bloesch, J. (2004). Sedimentation and lake sediment formation. In P. E. O'Sullivan & C. S.
 Reynolds (Eds.), *The lakes handbook. Volume 1. Limnology and limnetic ecology* (pp. 197-229). Oxford (UK): Blackwell. https://doi.org/10.1002/9780470999271.ch8
- Boltovskoy, D. (Ed.) (2015). *Limnoperna fortunei: the ecology, distribution and control of a swiftly spreading invasive fouling mussel*. Cham (Switzerland): Springer International Publishing. https://doi.org/10.1007/978-3-319-13494-9
- Boltovskoy, D., & Correa, N. (2015). Ecosystem impacts of the invasive bivalve *Limnoperna fortunei* (golden mussel) in South America, *Hydrobiologia*, *746*(1), 81-95. https://doi.org/10.1007/s10750-014-1882-9
- Boltovskoy, D., Xu, M., & Nakano, D. (2015a). Impacts of *Limnoperna fortunei* on man-made structures and control strategies: general overview. In D. Boltovskoy

(Ed.)*Limnoperna fortunei: the ecology, distribution and control of a swiftly spreading invasive fouling mussel* (pp. 375-393). Cham (Switzerland): Springer International Publishing. https://doi.org/10.1007/978-3-319-13494-9_21

Boltovskoy, D., Correa, N., Sylvester, F., & Cataldo, D. (2015b). Nutrient recycling, phytoplankton grazing, and associated impacts of *Limnoperna fortunei*. In D. Boltovskoy (Ed.)*Limnoperna fortunei: the ecology, distribution and control of a swiftly*

spreading invasive fouling mussel (pp. 153-176). Cham (Switzerland): Springer International Publishing. https://doi.org/10.1007/978-3-319-13494-9_9

Boltovskoy, D., Karatayev, A. Y., Burlakova, L. E., Cataldo, D., Karatayev, V., Sylvester, F., & Mariñelarena, A. (2009). Significant ecosystem-wide effects of the swiftly spreading

invasive freshwater bivalve *Limnoperna fortunei*, *Hydrobiologia*, *636*(1), 271-284. https://doi.org/10.1007/s10750-009-9956-9

- Bonachea, J., et al. (2010). Natural and human forcing in recent geomorphic change; case studies in the Rio de la Plata basin, *Science of the Total Environment*, *408*(13), 2674-2695. https://doi.org/10.1016/j.scitotenv.2010.03.004
- Boyd, A. J., & Heasman, K. G. (1998). Shellfish mariculture in the Benguela system: water flow patterns within a mussel farm in Saldanha Bay, South Africa, *Journal of Shellfish Research*, *17*(1), 25-32
- Burlakova, L. E., Karatayev, A. Y., & Karatayev, V. A. (2012). Invasive mussels induce community changes by increasing habitat complexity, *Hydrobiologia*, *685*(1), 121-134. https://doi.org/10.1007/s10750-011-0791-4
- Carlsson, M. S., Holmer, M., & Petersen, J. K. (2009). Seasonal and spatial variations of benthic impacts of mussel longline farming in a eutrophic Danish fjord, Limfjorden, *Journal of Shellfish Research*, 28(4), 791-801. https://doi.org/10.2983/035.028.0408
- Cataldo, D., O'Farrell, I., Paolucci, E. M., Sylvester, F., & Boltovskoy, D. (2012a). Impact of the invasive golden mussel (*Limnoperna fortunei*) on phytoplankton and nutrient cycling, *Aquatic Invasions*, 7(1), 91-100. https://doi.org/10.3391/ai.2012.7.1.010
- Cataldo, D., Vinocur, A., O'Farrell, I., Paolucci, E. M., Leites, V., & Boltovskoy, D. (2012b). The introduced bivalve *Limnoperna fortunei* boosts *Microcystis* growth in Salto Grande Reservoir (Argentina): evidence from mesocosm experiments, *Hydrobiologia*, *680*(1), 25-38. https://doi.org/10.1007/s10750-011-0897-8
- Chaffin, J. D., & Kane, D. D. (2010). Burrowing mayfly (Ephemeroptera: Ephemeridae: *Hexagenia* spp.) bioturbation and bioirrigation: A source of internal phosphorus loading in Lake Erie, *Journal of Great Lakes Research*, *36*(1), 57-63. https://doi.org/10.1016/j.jglr.2009.09.003
- Cicchetti, G., Latimer, J. S., Rego, S. A., Nelson, W. G., Bergen, B. J., & Coiro, L. L. (2006). Relationships between near-bottom dissolved oxygen and sediment profile camera measures, *Journal of Marine Systems*, *62*(3), 124-141.
 - https://doi.org/10.1016/j.jmarsys.2006.03.005
- Colombo, J. C., Cappelletti, N., Speranza, E., Migoya, M. C., Lasci, J., & Skorupka, C. N. (2007). Vertical fluxes and organic composition of settling material from the sewage impacted Buenos Aires coastal area, Argentina, *Organic Geochemistry*, *38*(11), 1941–1952. https://doi.org/10.1016/j.orggeochem.2007.07.005
- Crawford, C. M., Macleod, C. K. A., & Mitchell, I. M. (2003). Effects of shellfish farming on the benthic environment, *Aquaculture*, *224*(1-4), 117-140. https://doi.org/10.1016/S0044-8486(03)00210-2

- Dame, R. F., & Olenin, S. (Eds.) (2005). *The comparative roles of suspension-feeders in ecosystems*. Dodrecht (The Netherlands): Springer.
- Depetris, P. J., Gaiero, D. M., Probst, J. L., Hartmann, J., & Kempe, S. (2005). Biogeochemical output and typology of rivers draining Patagonia's Atlantic seaboard, *Journal of Coastal Research*, *21*(4), 835-844. https://doi.org/10.2112/015-nis.1
- Di Gregorio, D. E., Fernández Niello, J. O., Huck, H., Somacal, H., & Curutchet, G. (2007). ²¹⁰Pb dating of sediments in a heavily contaminated drainage channel to the La Plata estuary in Buenos Aires, Argentina, *Applied Radiation and Isotopes*, *65*(1), 126-130. https://doi.org/10.1016/j.apradiso.2006.06.008
- Dillon, P. J., Evans, R. D., & Molot, L. A. (1990). Retention and resuspension of phosphorus, nitrogen, and iron in a central Ontario lake, *Canadian Journal of Fisheries and Aquatic Sciences*, *47*(7), 1269-1274. https://doi.org/10.1139/f90-145
- Framiñan, M. B., Etala, M. P., Acha, E. M., Guerrero, R. A., Lasta, C. A., & Brown, O. B. (1999). Physical characteristics and processes of the Río de la Plata Estuary. In G. M. E. Perillo, M. C. Piccolo & M. Pino-Quivira (Eds.), *Estuaries of South America. Their geomorphology and dynamics* (pp. 161-194). Berlin (Germany): Springer. https://doi.org/10.1007/978-3-642-60131-6_8
- Gergs, R., Rinke, K., & Rothhaupt, K. O. (2009). Zebra mussels mediate benthic-pelagic coupling by biodeposition and changing detrital stoichiometry, *Freshwater Biology*, 54(7), 1379-1391. https://doi.org/10.1111/j.1365-2427.2009.02188.x
- Graf, G., & Rosenberg, R. (1997). Bioresuspension and biodeposition: a review, *Journal of Marine Systems*, *11*(3-4), 269-278. https://doi.org/10.1016/S0924-7963(96)00126-1
- Grenz, C., Hermin, M.-N., Baudinet, D., & Daumas, R. (1990). In situ biochemical and bacterial variation of sediments enriched with mussel biodeposits, *Hydrobiologia*, 207(1), 153-160. https://doi.org/10.1007/BF00041452
- Hammer, Ø., Harper, D. A. T., & Ryan, P. D. (2001). PAST: Paleontological statistics
 software package for education and data analysis, *Palaeontologia Electronica*, *4*, 1-9.
 Hartstein, N. D., & Rowden, A. A. (2004). Effect of biodeposits from mussel culture on
- macroinvertebrate assemblages at sites of different hydrodynamic regime, *Marine* Environmental Research, 57(5), 339-357.
 - https://doi.org/10.1016/j.marenvres.2003.11.003
- Hatcher, A., Grant, J., & Schofield, B. (1994). Effects of suspended mussel culture (*Mytilus* spp.) on sedimentation, benthic respiration and sediment nutrient dynamics in a coastal bay, *Marine Ecology Progress Series*, *115*(3), 219-235. https://doi.org/10.3354/meps115219
- Hoogsteen, M. J. J., Lantinga, E. A., Bakker, E. J., Groot, J. C. J., & Tittonell, P. A. (2015). Estimating soil organic carbon through loss on ignition: effects of ignition conditions

and structural water loss, *European Journal of Soil Science*, *66*(2), 320-328. https://doi.org/10.1111/ejss.12224

- Horppila, J., & Nurminen, L. (2005). Effects of different macrophyte growth forms on sediment and P resuspension in a shallow lake, *Hydrobiologia*, *545*(1), 167-175. https://doi.org/10.1007/s10750-005-2677-9
- Jordan, S. J. (1987). Sedimentation and remineralization associated with biodeposition by the American oyster *Crassostrea virginica* (Gmelin), PhD thesis, 200 pp., University of Maryland, College Park (USA)
- Jørgensen, C. B. (1990). *Bivalve filter feeding: hydrodynamics, bioenergetics, physiology* and ecology. Fredensborg (Denmark): Olsen and Olsen.
- Karatayev, A. Y., Burlakova, L. E., Karatayev, V. A., & Boltovskoy, D. (2010). *Limnoperna fortunei* versus *Dreissena polymorpha*: Population densities and benthic community impacts of two invasive freshwater bivalves, *Journal of Shellfish Research*, 29(4), 975-984. https://doi.org/10.2983/035.029.0432
- Karatayev, A. Y., Padilla, D. K., Minchin, D., Boltovskoy, D., & Burlakova, L. E. (2007).
 Changes in global economies and trade: the potential spread of exotic freshwater bivalves, *Biological Invasions*, *9*(2), 161-180. https://doi.org/10.1007/s10530-006-9013-9
- Karatayev, A. Y., Burlakova, L. E., Padilla, D. K., Mastitsky, S. E., & Olenin, S. (2009). Invaders are not a random selection of species, *Biological Invasions*, *11*(9), 2009-2019. https://doi.org/10.1007/s10530-009-9498-0
- Kaspar, H. F., Gillespie, P. A., Boyer, I. C., & MacKenzie, A. L. (1985). Effects of mussel aquaculture on the nitrogen cycle and benthic communities in Kenepuru Sound, Marlborough Sounds, New Zealand, *Marine Biology*, *85*(2), 127-136. https://doi.org/10.1007/BF00397431
- Kleeberg, A. (2002). Phosphorus sedimentation in seasonal anoxic Lake Scharmützel, NE
 Germany, *Hydrobiologia*, 472(1), 53-65. https://doi.org/10.1023/A:1016356714276
 Klerks, P. L., Fraleigh, P. C., & Lawniczak, J. E. (1996). Effects of zebra mussel (*Dreissena*)
 - *polymorpha*) on seston levels and sediment deposition in western Lake Erie, *Canadian Journal of Fisheries and Aquatic Sciences*, *53*(10), 2284-2291. https://doi.org/10.1139/f96-190
- Li, H., Liu, L., Li, M., & Zhang, X. (2013). Effects of pH, temperature, dissolved oxygen, and flow rate on phosphorus release processes at the sediment and water interface in storm sewer, *Journal of Analytical Methods in Chemistry*, 2013, 104316. https://doi.org/10.1155/2013/104316

- Linares, M. S., Callisto, M., & Marques, J. C. (2017). Invasive bivalves increase benthic communities complexity in neotropical reservoirs, *Ecological Indicators*, *75*, 279-285. https://doi.org/10.1016/j.ecolind.2016.12.046
- MacIsaac, H. J., & Rocha, R. (1995). Effects of suspended clay on zebra mussel (*Dreissena polymorpha*) faeces and pseudofaeces production, *Archiv für Hydrobiologie*, *135*(1), 53-64.
- Mayer, C. M., et al. (2014). Benthification of freshwater lakes. Exotic mussels turning
 ecosystems upside down. In T. F. Nalepa & D. W. Schloesser (Eds.), *Quagga and* zebra mussels: biology, impacts, and control. Second edition (pp. 575-585). Boca
 Raton (USA): CRC Press. https://doi.org/10.1201/b15437-44
- Mazouni, N., Gaertner, J. C., Deslous-Paoli, J. M., Landrein, S., & Geringer d'Oedenberg, M. (1996). Nutrient and oxygen exchanges at the water sediment interface in a shellfish farming lagoon (Thau, France), *Journal of Experimental Marine Biology and Ecology*, 205(1-2), 92-113. https://doi.org/10.1016/S0022-0981(96)02594-4
- McCall, P. L., Matisoff, G., & Tevesz, M. J. S. (1986). The effects of a unionid bivalve on the physical, chemical, and microbial properties of cohesive sediments from Lake Erie, *American Journal of Science*, *286*(2), 127-159. https://doi.org/10.2475/ajs.286.2.127
- Mellina, E., Rasmussen, J. B., & Mills, E. L. (1995). Impact of zebra mussel (*Dreissena polymorpha*) on phosphorus cycling and chlorophyll in lakes, *Canadian Journal of Fisheries and Aquatic Sciences*, *52*(12), 2552-2573. https://doi.org/10.1139/f95-246
- Mida, J. L., Scavia, D., Fahnenstiel, G. L., Pothoven, S. A., Vanderploeg, H. A., & Dolan, D.
 M. (2010). Long-term and recent changes in southern Lake Michigan water quality with implications for present trophic status, *Journal of Great Lakes Research*, *36*(3), 42-49. https://doi.org/10.1016/j.jglr.2010.03.010
- Moreira, D., Simionato, C. G., Gohin, F., Cayocca, F., & Tejedor, M. L. C. (2013). Suspended matter mean distribution and seasonal cycle in the Río de La Plata estuary and the adjacent shelf from ocean color satellite (MODIS) and in-situ observations, *Continental Shelf Research*, *68*, 51-66. https://doi.org/10.1016/j.csr.2013.08.015
- Moreira, D., Simionato, C. G., Dragani, W., Cayocca, F., & Tejedor, M. L. C. (2016). Characterization of bottom sediments in the Río de la Plata estuary, *Journal of Coastal Research*, *32*(6), 1473-1494. https://doi.org/10.2112/JCOASTRES-D-15-00078.1
- Mosley, C., & Bootsma, H. (2015). Phosphorus recycling by profunda quagga mussels (*Dreissena rostriformis bugensis*) in Lake Michigan, *Journal of Great Lakes Research, 41*(3), 38-48. https://doi.org/10.1016/j.jglr.2015.07.007

- Musin, G. E., Rojas Molina, F., Giri, F., & Williner, V. (2015). Structure and density population of the invasive mollusc *Limnoperna fortunei* associated with *Eichhornia crassipes* in lakes of the Middle Paraná floodplain, *Journal of Limnology*, *74*(3), 537-548. https://doi.org/10.4081/jlimnol.2015.1107
- Nalepa, T. F., & Schloesser, D. W. (Eds.) (2014). *Quagga and zebra mussels: biology, impacts, and control. Second Edition.* Boca Raton (USA): CRC Press.
- Nalepa, T. F., Gardner, W. S., & Malczyk, J. M. (1991). Phosphorus cycling by mussels (Unionidae: Bivalvia) in Lake St. Clair, *Hydrobiologia*, *219*(1), 239-250. https://doi.org/10.1007/BF00024758
- Newell, R. I. E. (2004). Ecosystem influences of natural and cultivated populations of suspension-feeding bivalve molluscs: a review, *Journal of Shellfish Research*, *23*(1), 51-61.
- Newell, R. I. E., Fisher, T. R., Holyoke, R. R., & Cornwell, J. C. (2005). Influence of eastern oysters on nitrogen and phosphorus regeneration in Chesapeake Bay, USA. In R. F. Dame & S. Olenin (Eds.), *The Comparative Roles of Suspension-Feeders in Ecosystems: Proceedings of the NATO Advanced Research Workshop on The Comparative Roles of Suspension-Feeders in Ecosystems Nida, Lithuania 4–9 October 2003* (pp. 93-120). Dordrecht (Netherlands): Springer. https://doi.org/10.1007/1-4020-3030-4_6
- Oliveira, M. D., Campos, M. C. S., Paolucci, E. M., Mansur, M. C. D., & Hamilton, S. K.
 (2015). Colonization and spread of *Limnoperna fortunei* in South America. In D.
 Boltovskoy (Ed.)*Limnoperna fortunei: the ecology, distribution and control of a swiftly spreading invasive fouling mussel* (pp. 333-355). Cham (Switzerland): Springer
 International Publishing. https://doi.org/10.1007/978-3-319-13494-9_19
- Ozersky, T., Evans, D. O., & Ginn, B. K. (2015). Invasive mussels modify the cycling, storage and distribution of nutrients and carbon in a large lake, *Freshwater Biology*, *60*(4), 827-843. https://doi.org/10.1111/fwb.12537
- Reeders, H. H., & bij de Vaate, A. (1992). Bioprocessing of polluted suspended matter from the water column by the zebra mussel (*Dreissena polymorpha* Pallas), *Hydrobiologia*, 239(1), 53-63. https://doi.org/10.1007/BF00027529
- Reynolds, C. S., & Davies, P. S. (2001). Sources and bioavailability of phosphorus fractions in freshwaters: A British perspective, *Biological Reviews of the Cambridge Philosophical Society*, *76*(1), 27-64. https://doi.org/10.1111/j.1469-185X.2000.tb00058.x
- Roditi, H. A., Strayer, D. L., & Findlay, S. E. G. (1997). Characteristics of zebra mussel (*Dreissena polymorpha*) biodeposits in a tidal freshwater estuary, *Archiv für*

Hydrobiologie, *140*(2), 207-219. https://doi.org/10.1127/archivhydrobiol/140/1997/207

- Ronco, A., Peluso, L., Jurado, M., Bulus Rossini, G., & Salibian, A. (2008). Screening of sediment pollution in tributaries from the southwestern coast of the Río de la Plata estuary, *Latin American Journal of Sedimentology and Basin Analysis*, *15*(1), 67-75.
- Ruginis, T., Bartoli, M., Petkuviene, J., Zilius, M., Lubiene, I., Laini, A., & Razinkovas-Baziukas, A. (2014). Benthic respiration and stoichiometry of regenerated nutrients in lake sediments with *Dreissena polymorpha*, *Aquatic Sciences*, *76*(3), 405-417. https://doi.org/10.1007/s00027-014-0343-x
- Sardiña, P., Cataldo, D., & Boltovskoy, D. (2008). The effects of the invasive mussel, *Limnoperna fortunei*, on associated fauna in South American freshwaters: importance of physical structure and food supply, *Fundamental and Applied Limnology / Archiv für Hydrobiologie*, *173*(2), 135-144. https://doi.org/10.1127/1863-9135/2008/0173-0135
- Sardiña, P., Chaves, E., & Marchese, M. (2011). Benthic community responses to invasion by the golden mussel, *Limnoperna fortunei* Dunker: biotic homogenization vs environmental driving forces, *Journal of the North American Benthological Society*, *30*(4), 1009-1023. https://doi.org/10.1899/10-170.1
- Sarubbi, A. (2007). Análisis del avance del frente del Delta del Río Paraná, PhD thesis, 136 pp., Universidad de Buenos Aires (Argentina)
- Schuerch, M., Scholten, J., Carretero, S., García-Rodríguez, F., Kumbier, K., Baechtiger, M.,
 & Liebetrau, V. (2016). The effect of long-term and decadal climate and hydrology
 variations on estuarine marsh dynamics: An identifying case study from the Río de la
 Plata, *Geomorphology*, 269, 122-132.
 - https://doi.org/10.1016/j.geomorph.2016.06.029
- Shumway, S. E. (Ed.) (2011). Shellfish aquaculture and the environment. Ames (USA): Wiley-Blackwell. https://doi.org/10.1002/9780470960967
- Simionato, C. G., Moreira, D., Re, M., & Fossati, M. (2011). Estudio de la dinámica hidrosedimentológica del Río de la Plata: observación y modelación numérica de los sedimentos finos, *Rep. Proyecto FREPLATA,CTMFM-CARP (Argentina-Uruguay)*, 1-108 pp.
- Smyth A. R., Piehler M. F., Grabowski, J. H., & Frid, C. (2015). Habitat context influences nitrogen removal by restored oyster reefs, *Journal of Applied Ecology*, *52*(3), 716-725. https://doi.org/10.1111/1365-2664.12435
- Sobczyński, T., & Joniak, T. (2009). Vertical changeability of physical-chemical features of bottom sediments in three lakes, in aspect type of water mixis and intensity of human impact, *Polish Journal of Environmental Studies*, *18*(9), 1093-1099.

- Sornin, J. M., Feuillet, M., M., H., & Fardeau, J. C. (1986). Influence des cultures d'huîtres *Crassostrea gigas* sur le cycle du phosphore en zone irltertidale: rôle de la biodéposition, *Oceanologica Acta*, *9*(3), 313-322.
- Sprung, M., & Rose, U. (1988). Influence of food size and food quantity on the feeding of the mussel *Dreissena polymorpha*, *Oecologia*, 77(4), 526-532.

https://doi.org/10.1007/BF00377269

- Strayer, D. L. (2014). Understanding how nutrient cycles and freshwater mussels (Unionoida) affect one another, *Hydrobiologia*, 735(1), 277-292. https://doi.org/10.1007/s10750-013-1461-5
- Sylvester, F., & Sardiña, P. (2015). Relationships of *Limnoperna fortunei* with benthic animals. In D. Boltovskoy (Ed.)*Limnoperna fortunei: the ecology, distribution and control of a swiftly spreading invasive fouling mussel* (pp. 191-210). Cham (Switzerland): Springer International Publishing. https://doi.org/10.1007/978-3-319-13494-9_11
- Sylvester, F., Boltovskoy, D., & Cataldo, D. (2007a). Fast response of freshwater consumers to a new trophic resource: Predation on the recently introduced Asian bivalve *Limnoperna fortunei* in the lower Paraná River, South America, *Austral Ecology*, 32(4), 403-415. https://doi.org/10.1111/j.1442-9993.2007.01707.x
- Sylvester, F., Boltovskoy, D., & Cataldo, D. (2007b). The invasive bivalve *Limnoperna* fortunei enhances benthic invertebrate densities in South American floodplain rivers, *Hydrobiologia*, *589*(1), 15-27. https://doi.org/10.1007/s10750-007-0708-4
- Sylvester, F., Dorado, J., Boltovskoy, D., Juárez, A., & Cataldo, D. (2005). Filtration rates of the invasive pest bivalve *Limnoperna fortunei* as a function of size and temperature, *Hydrobiologia*, *534*(1-3), 71-80. https://doi.org/10.1007/s10750-004-1322-3
- Tokumon, R., Cataldo, D., & Boltovskoy, D. (2015). Effects of suspended inorganic matter on filtration and grazing rates of the invasive mussel *Limnoperna fortunei* (Bivalvia, Mytiloidea), *Journal of Molluscan Studies*, *8*2(1), 201-204. https://doi.org/10.1093/mollus/eyv024
- Torres, I. C., Inglett, P. W., Brenner, M., Kenney, W. F., & Ramesh Reddy, K. (2012). Stable isotope (δ¹³C and δ¹⁵N) values of sediment organic matter in subtropical lakes of different trophic status, *Journal of Paleolimnology*, *47*(4), 693-706. https://doi.org/10.1007/s10933-012-9593-6
- Tye, A. M., Rawlins, B. G., Rushton, J. C., & Price, R. (2016). Understanding the controls on sediment-P interactions and dynamics along a non-tidal river system in a rural–urban catchment: the River Nene, *Applied Geochemistry*, *66*, 219-233. https://doi.org/10.1016/j.apgeochem.2015.12.014

- Valderrama, J. C. (1981). The simultaneous analysis of total nitrogen and total phosphorus in natural waters, *Marine Chemistry*, *10*(2), 109–122. https://doi.org/10.1016/0304-4203(81)90027-X
- Vanderploeg, H. A., Sarnelle, O., Liebig, J. R., Morehead, N. R., Robinson, S. D., Johengen,
 T. H., & Horst, G. P. (2017). Seston quality drives feeding, stoichiometry and
 excretion of zebra mussels, *Freshwater Biology*, *62*(4), 664-680.
 https://doi.org/10.1111/fwb.12892
- Wetzel, R. G. (2001). *Limnology: Lake and river ecosystems*. San Diego (USA): Academic Press.
- Yamada, Y., Mito, Y., Igeta, A., & Wada, E. (2012). Dissolved oxygen concentration in river sediment of the Lake Biwa tributaries, Japan, *Limnology*, *13*(1), 149-154. https://doi.org/10.1007/s10201-011-0348-2
- Zaiko, A., Paskauskas, R., & Krevs, A. (2010). Biogeochemical alteration of the benthic environment by the zebra mussel *Dreissena polymorpha* (Pallas), *Oceanologia*, *52*(4), 649-667. https://doi.org/10.5697/oc.52-4.649

Accepted



Figure 1. General scheme of the experimental setup (a), and code numbers for the experimental units (b).

Accept



Figure 2. Schematic detail of experimental units with mussels (+M, a), and without mussels (controls, CTR, b). c: Perspective view and measurements (in mm) of the experimental units.



Figure 3. Mean and SD of accumulated sediments (a) and % of organic matter (OM) in the sediments (b) in the experimental units. For OM, bimonthly and annual data are means of all 2 cm intervals in the cores obtained from these EU (Figure 6). See Table S1 in the supporting information for data and units.





way ANOVA and Tukey's contrasts; KW: Kruskal-Wallis test and Mann-Whitney contrasts). See Table S1 in the supporting information for data and units.

Acc



Figure 5. Relationships between the variables assessed in CTR vs. +M. Regression lines and R and P values are indicated for the significant (P<0.05) correlations. All datapoints and regression lines are based on actual values, correlation R and P are Pearson's (P) or Spearman's (S; when either variable did not comply with normality). a, b: linear; d: power. See Table S1 in the supporting information for data and units.



Figure 6. Mean and SD values for % organic matter (a-c), total P (d-f) and total N (g-i) in the cores retrieved from the two biannual and the annual deployments at 2 cm intervals. Each value is based on 2-6 replicates. Values with mussels (+M) are normalized to the average mussel biomass for all experimental units (5.2 g of mussel tissue dry weight). Notice that experiments with mussels accumulated more sediments, and therefore their cores are longer than those of the controls (CTR).

Accept



Figure 7. Mean and SD values for total P (a), and total N (b) in sediments accumulated in the experimental units. Bimonthly and annual data are means of all 2 cm intervals in the cores obtained from the EU (Figure 6). Difference between CTR and +M means, total P: P=0.482; total N: P=0.002 (Mann-Whitney tests). See Table S1 in the supporting information for data and units.

Ac